**Author’s response to reviews**

**Title:** Establishing spatially-enabled health registry systems using implicit spatial data pools: Case study – Uganda

**Authors:**
Augustus Aturinde (augustus.aturinde@nateko.lu.se)
Rose Nakasi (g.nakasi.rose@gmail.com)
Mahdi Farnaghi (mahdi.farnaghi@nateko.lu.se)
Gilbert Maiga (gilmaiga@gmail.com)
Petter Pilesjo (Petter.pilesjo@gis.lu.se)
Ali Mansourian (ali.mansourian@nateko.lu.se)

**Version:** 3 **Date:** 17 Sep 2019

**Author’s response to reviews:**

**Editor Comments:**
Based on these reports, and my own assessment as Editor, I am pleased to inform you that it is potentially acceptable for publication in BMC Medical Informatics and Decision Making, once you have carried out some essential revisions suggested by our reviewers.

**Response:**
Thank you for finding our work acceptable.

**Comment 1**
Could you please either delete Figure 4 or remove all patient identifiable data (such as name, phone number etc)

**Response:**
Figure 4 has been deleted and the text edited accordingly.

**Comment 2**
Could I please ask from where you have sourced the map images for figure 5 and indicate whether you have permissions to use them?

**Response:**
Figure 5 was produced from the sample data the authors collected with the use of the mobile
application (Background map from ArcGIS-Online). A text indicating the same has been included in the Figure 5 legend.

Comment 3
Please remove all figure captions from the figure files and move all Figure legends to after the references section

Response:
Implemented

Comment 4
Please rename "Introduction" section to "Background" section. Please upload a clean manuscript file with no highlighting or tracking changes

Response:
Implemented